Author's response to reviews

Title: Consumption of Cisatracurium in different age groups, using a closed loop computer controlled system

Authors:

Shehzaad Joomye (s_joomye@hotmail.com)
Guolin Wang (wang_quolin@hotmail.com)
Haiyun Wang (wanghy819@hotmail.com)
Donglei Yan (yyvddl@sina.com)
Guoqiang Zhou (mzkzgq@163.com)

Version: 5
Date: 12 March 2014

Author's response to reviews: see over
Dear Editors and reviewers,

Thank you very much for your careful review and constructive suggestions with regard to our manuscript “Consumption of Cisatracurium in different age groups, using a closed loop computer controlled system”.

Those comments are helpful for authors to revise and improve our paper. We have studied comments carefully and tried our best to revise and improve the manuscript and made great changes in the manuscript according to the referees’ good comments. Revised portion is marked in red in the paper. We appreciate for Editors/Reviewers’ warm work earnestly, and hope that the corrections will meet with approval. Please feel free to contact us with any questions and we are looking forward to your consideration. The main corrections in the paper and the responds to the reviewer’s comments are as following:

Responds to the reviewer’s comments:

Reviewer #3:
1. Response to comment: Please include an Authors’ Contributions section in your manuscript.

Response: We are very sorry for our negligence. An Authors’ Contributions has been added after an Acknowledgement section.

Special thanks to you for your good comments.

In all, we found that the reviewers’ comments are quite helpful, and we revised our paper point-by-point. Once again, thank you for your careful review and constructive suggestions!

The manuscript has been resubmitted to your journal. We are convinced that several aspects of this manuscript will make it interesting to general readers of BMC Anesthesiology. It has not been submitted for publication nor has it been published in whole or in part elsewhere. I attest to the fact that all authors listed on the title page have read the manuscript, attest to the validity and legitimacy of the data and its interpretation, and agree to its submission to BMC Anesthesiology. I acknowledge that both I and the other authors have read the Instructions for Authors and agree with its contents. I acknowledge that if the enclosed manuscript is part of a larger whole or
if the primary analysis has been previously published, this must be explicitly stated in
the manuscript and the previous publication cited.

We certified that none of the authors have any financial or scientific conflict of
interest with regard to the research described in this manuscript. I promise if accepted,
this article shall not be published elsewhere in the same form, in either the same or
another language, without the written consent of the Editor and Publisher of BMC
Anesthesiology. We look forward to your positive response.

Sincerely yours,
Guolin Wang
Department of Anesthesia,
Tianjin Medical University General Hospital
Tianjin Research Institute of Anesthesiology
Tianjin, P.R.China